Author's response to reviews

Title: Implication of Plasma Intermedin Levels In Patients Who Underwent First-time Diagnostic Coronary Angiography: a single centre, cross-sectional study

Authors:

Aylin Hatice Yamac (aylin-yamac@yandex.com)
Ahmet Bacaksiz (ahmetbacaxiz@yandex.com)
Ziya Ismailoglu (ziyaismail@yahoo.com)
Sitki Kucukbuzcu (s_kucukbuzcu@hotmail.com)
Emrah Sevgili (sevgiliemrah@yahoo.com)
Emin Asoglu (asemctff@hotmail.com)
Muharrem Nasifov (drmeherrem@yahoo.com)
Parviz Jaferov (doctorperviz@gmail.com)
Ercan Erdogan (drercanerdogan@gmail.com)
Omer Goktekin (ogoktekin@gmail.com)

Version: 4 Date: 17 November 2014

Author's response to reviews: see over
Dear Editor,

The authors have revised the manuscript entitled "Implication of Plasma Intermedin Levels In Patients Who Underwent First-time Diagnostic Coronary Angiography: a single centre, cross-sectional study" and have responded to the reviewers’ comments.

We are looking forward to your favorable response in your early convenience.

Yours sincerely,

Aylin Yamac, M.D.
Responses to Reviewers’ Comments:

Comment: "Please update your ethics statement to include the name of the ethics committee that approved your study."

Reply: We added the name of the local ethics committee into the manuscript as "The local ethics committee (BezmiÂlem Foundation University Clinical Studies Ethics Committee) approved the study protocol" according to the reviewer’s suggestions.

Comment: "Please provide an abstract formatted according to the guidelines for authors <http://www.biomedcentral.com/info/ifora/abstracts>. Potential referees will be asked to review the manuscript having seen only the title and abstract, so it is important that these are both informative and concise.

Please note that the abstract in your manuscript file differs from the abstract entered in the submission system. - Please correct so they are consistent with each other.

The abstract of the manuscript should not exceed 350* words and must be structured into separate sections:

Abstract sections

- Background

- Methods

- Results

- Conclusions"

Reply: We condensed the abstract according to the reviewer’s suggestions.